Author's response to reviews

Title: Outcomes for patients with advanced medical illness: a multicentre study of validity and predictors of response to change

Authors:

Massimo Costantini (costantini.massimo@asmn.re.it)
Elisa Rabitti (elisa.rabitti@ausl.re.it)
Monica Beccaro (monica.beccaro@asmepa.org)
Flavio Fusco (flavio.fusco@asl3.liguria.it)
Carlo Peruselli (carlo.peruselli@gmail.com)
Pietro La Ciura (laciura.pietro@alice.it)
Alessandro Valle (alessandro.valle@tin.it)
Cinzia Suriani (cinziasuri@alice.it)
Marco Maltoni (ma.maltoni@ausl.fo.it)
Danila Valenti (d.valenti@ausl.bo.it)
Felicità Mosso (fmosso@libero.it)
Piero Morino (piero.morino@asf.toscana.it)
Giovanni Zaninetta (zaninetta@gmail.com)
Giorgio Tubere (g.tubere@asl1.liguria.it)
Massimo Piazza (mpiazza52@gmail.com)
Michele Sofia (msofia@aogarbagnate.lombardia.it)
Silvia Di Leo (silvia.dileo@asmn.re.it)
Irene J Higginson (irene.higginson@kcl.ac.uk)

Version: 3
Date: 13 April 2015

Author's response to reviews: see over
Dear Editor,

please find enclosed our original article ‘Outcomes for patients with advanced medical illness: a multicentre study of validity and predictors of response to change’, that we are submitting for publication in ‘BMC Palliative Care’.

We have changed the manuscript according to your suggestions to make it suitable for external review. More specifically:
- we have included line and page numbers;
- we have included a “List of abbreviations” section.
- during the last revision we have realized that a mistake occurred in the list of authors. We have modified the list.

The article has not been published elsewhere and it is not being considered elsewhere for publication. All authors agreed to the content of the article and to its submission to ‘BMC Palliative Care’, and declared, as reported in a specific section of the text, they have no competing interests.

With best regards
Yours sincerely

Massimo Costantini, MD
Palliative Care Unit,
IRCCS Arcispedale S. Maria Nuova, Reggio Emilia (Italy)
Email: costantini.massimo@asmn.re.it